Author’s response to reviews

Title: National Mesothelioma Virtual Bank: a standard based biospecimen and clinical data resource to enhance translational research

Authors:

Waqas Amin (aminw@upmc.edu)
Anil V Parwani (parwaniav@upmc.edu)
Linda Schmandt (schmandtlm@upmc.edu)
Sambit K Mohanty (mohantys2@upmc.edu)
Ghada Farhat (farhatg@upmc.edu)
Andrew K Pople (popleak@upmc.edu)
Sharon B Winters (wintersssb@upmc.edu)
Nancy B Whelan (whelannb@upmc.edu)
Althea M Schneider (schneideram2@upmc.edu)
John T Milnes (milnesjt@upmc.edu)
Federico A Valdivieso (Federico.Valdivieso@uphs.upenn.edu)
Michael Feldman (feldmanm@mail.med.upenn.edu)
Harvey I Pass (harvey.pass@med.nyu.edu)
Rajiv Dhir (dhirr@upmc.edu)
Jonathan Melamed (jonathan.melamed@med.nyu.edu)
Michael J Becich (becich@pitt.edu)

Version: 6 Date: 11 August 2008

Author's response to reviews:

Revised Manuscript: 4677143731958176
National Mesothelioma Virtual Bank: a standard based biospecimen and clinical data resource to enhance translational research

I corrected some typographical errors. After submitting this revised version of my manuscript I approved it for publication.